Author’s response to reviews

Title: Self-competence in death work among health and social care workers: A region-wide survey in Hong Kong

Authors:
Johnny Cheung (johnnycheung@cuhk.edu.hk)
Doreen W. H. Au (doreenau@cuhk.edu.hk)
Wallace C. H. Chan (chchan@swk.cuhk.edu.hk)
Jenny H. Y. Chan (hyanchan@cuhk.edu.hk)
Kenway Ng (kenwayng@cuhk.edu.hk)
Jean Woo (jeanwoowong@cuhk.edu.hk)

Version: 2 Date: 09 Apr 2018

Author’s response to reviews:

Thank you for completing the revisions to the manuscript. There was just one error in the Abstract in that you removed the sample size from the methods section but did not add it into the results section. It is important to include the sample size in the abstract.

- In Abstract Result Section, line 70, page 3, we added “We collected data from 885 health and social care workers.”